

THE UNIVERSITY of EDINBURGH

Edinburgh Research Explorer

General contextual effects on neglected tropical disease risk in rural Kenya

Citation for published version:

de Glanville, W, Thomas, LF, Cook, EAJ, Bronsvoort, M, Wardrop, N, Wamae, CN, Kariuki, S & Fevre, EM 2018, 'General contextual effects on neglected tropical disease risk in rural Kenya', *PLoS Neglected Tropical Diseases*, vol. 12, no. 12, e0007016. https://doi.org/10.1371/journal.pntd.0007016

Digital Object Identifier (DOI):

10.1371/journal.pntd.0007016

Link:

Link to publication record in Edinburgh Research Explorer

Document Version: Publisher's PDF, also known as Version of record

Published In: PLoS Neglected Tropical Diseases

Publisher Rights Statement:

Copyright: © 2018 de Glanville et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

General rights

Copyright for the publications made accessible via the Edinburgh Research Explorer is retained by the author(s) and / or other copyright owners and it is a condition of accessing these publications that users recognise and abide by the legal requirements associated with these rights.

Take down policy

The University of Édinburgh has made every reasonable effort to ensure that Edinburgh Research Explorer content complies with UK legislation. If you believe that the public display of this file breaches copyright please contact openaccess@ed.ac.uk providing details, and we will remove access to the work immediately and investigate your claim.





Citation: de Glanville WA, Thomas LF, Cook EAJ, Bronsvoort BMdC, Wardrop N, Wamae CN, et al. (2018) General contextual effects on neglected tropical disease risk in rural Kenya. PLoS Negl Trop Dis 12(12): e0007016. https://doi.org/10.1371/ journal.pntd.0007016

Editor: Jeremiah M. Ngondi, RTI International, UNITED REPUBLIC OF TANZANIA

Received: May 16, 2018

Accepted: November 22, 2018

Published: December 21, 2018

Copyright: © 2018 de Glanville et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Anonymised data described in this study are available at: 10.17638/ datacat.liverpool.ac.uk/448.

Funding: WAdeG and LFT were supported by BBSRC DTG awards, and EAJC by an MRC DTG award. EMF and the project described were supported by the Wellcome Trust (085308). This work was also supported by a Strategic Award from the Wellcome Trust to the Centre for Immunity, Infection and Evolution (095831), supporting the laboratory facilities in Busia. This **RESEARCH ARTICLE**

General contextual effects on neglected tropical disease risk in rural Kenya

William A. de Glanville^{1,2¤a}*, Lian F. Thomas^{1,2}, Elizabeth A. J. Cook^{1,2}, Barend M. de C. Bronsvoort³, Nicola Wardrop^{4¤b}, Claire N. Wamae⁵, Samuel Kariuki⁶, Eric M. Fèvre^{2,7}*

1 Centre for Immunity, Infection and Evolution, Institute for Immunology and Infection Research, School of Biological Sciences, University of Edinburgh, Edinburgh, United Kingdom, 2 International Livestock Research Institute, Nairobi, Kenya, 3 The Roslin Institute, Royal (Dick) School of Veterinary Studies, University of Edinburgh, Roslin, Midlothian, United Kingdom, 4 Department of Geography and Environment, University of Southampton, Highfield Campus, Southampton, United Kingdom, 5 School of Pharmacy and Health Sciences, United States International University, Nairobi, Kenya, 6 Centre for Microbiology Research, Kenya Medical Research Institute, Nairobi, Kenya, 7 Institute of Infection and Global Health, University of Liverpool, Leahurst Campus, Neston, United Kingdom

Pa Current address: Institute of Biodiversity, Animal Health and Comparative Medicine, University of Glasgow, Glasgow, United Kingdom,

pb Current address: Department for International Development, Abercrombie House, Eaglesham Road, East Kilbride, United Kingdom

* Will.deGlanville@glasgow.ac.uk (WAG); Eric.Fevre@liverpool.ac.uk (EMF)

Abstract

The neglected tropical diseases (NTDs) are characterized by their tendency to cluster within groups of people, typically the poorest and most marginalized. Despite this, measures of clustering, such as within-group correlation or between-group heterogeneity, are rarely reported from community-based studies of NTD risk. We describe a general contextual analysis that uses multi-level models to partition and quantify variation in individual NTD risk at multiple grouping levels in rural Kenya. The importance of general contextual effects (GCE) in structuring variation in individual infection with Schistosoma mansoni, the soil-transmitted helminths, Taenia species, and Entamoeba histolytica/dispar was examined at the household-, sublocation- and constituency-levels using variance partition/intra-class correlation co-efficients and median odds ratios. These were compared with GCE for HIV, Plasmodium falciparum and Mycobacterium tuberculosis. The role of place of residence in shaping infection risk was further assessed using the spatial scan statistic. Individuals from the same household showed correlation in infection for all pathogens, and this was consistently highest for the gastrointestinal helminths. The lowest levels of household clustering were observed for E. histolytica/dispar, P. falciparum and M. tuberculosis. Substantial heterogeneity in individual infection risk was observed between sublocations for S. mansoni and Taenia solium cysticercosis and between constituencies for infection with S. mansoni, Trichuris trichiura and Ascaris lumbricoides. Large overlapping spatial clusters were detected for S. mansoni, T. trichiura, A. lumbricoides, and Taenia spp., which overlapped a large cluster of elevated HIV risk. Important place-based heterogeneities in infection risk exist in this community, and these GCEs are greater for the NTDs and HIV than for TB and malaria. Our findings suggest that broad-scale contextual drivers shape infectious disease risk in this population, but these effects operate at different grouping-levels for different pathogens. A

work received support from the CGIAR Research Program on Agriculture for Nutrition and Health (A4NH), led by the International Food Policy Research Institute (IFPRI). We also acknowledge the CGIAR Fund Donors (http://www.cgiar.org/ about-us/our-funders/). The funders had no role in study design, analysis, manuscript preparation or decision to publish.

Competing interests: The authors have declared that no competing interests exist.

general contextual analysis can provide a foundation for understanding the complex ecology of NTDs and contribute to the targeting of interventions.

Author summary

Variation in infectious disease risk between groups of individuals represent health inequalities: reducing these inequalities, alongside reductions in infection prevalence, is a major focus for public health interventions. Despite this, it is rare that general contextual effects, or measures of within-group correlation or between-group heterogeneity, are reported as substantive outcomes from community-based studies of infectious disease risk, including for the NTDs. This reflects wider issues around a lack of social epidemiological perspectives, or consideration of the effects of contextual drivers, in communicable disease research, particularly in low-income settings. The aim of this study was to measure general contextual effects on human infection risk for a number of endemic helminth, protozoal, bacterial and viral pathogens in a rural farming community in western Kenya. Using this approach, we reveal clustering at a range of administrative and geographic levels and are able to show that the magnitude of clustering, and the hierarchical grouping level at which it occurs (from the household to administrative constituency), varies substantially between pathogens. Greater within-group correlation and between-group heterogeneity in infection risk was observed for the helminth NTDs and HIV than for Entamoeba histolytica/dispar, Plasmodium falciparum or Mycobacterium tuberculosis. Quantification of general contextual effects can inform the design of interventions that aim to reduce health inequalities within a population and can provide actionable targets for assessing the short- and long-term impact of interventions.

Introduction

People living in rural areas in sub-Saharan Africa are often at high risk of infection with a range of pathogens [1-3]. The burden of preventable infectious disease in many of these communities can perpetuate poverty [4], reduce well-being [5,6], and contribute to high rates of mortality [7]. An individual's risk of infection with any pathogen depends on a complex interplay of factors that relate to their exposure and susceptibility [8]. The individual-level characteristics that determine the likelihood of encountering a particular pathogen, and of infection following exposure, are often greatly influenced by the social, cultural, political, economic and/or environmental contextual conditions in which a person lives [9-11]. Since individuals living in the same geographic, administrative or institutional setting are generally exposed to the same contextual conditions (although not necessarily in the same way), adverse health outcomes commonly cluster within particular grouping levels. Hence, all else being equal, two people living in different groups [12]. Such clustering effects are often large for infectious diseases, and particularly so at the household-level for pathogens that are spread through poor sanitation, contaminated water, endophagic vectors, and unhygienic practices [13–18].

Clustering of infection within groups, and the contextual effects that drive it, such as marginalization, poverty and access to health services, is integral to the conceptualization of an infectious disease as 'neglected' [19]. However, it has been suggested that effects acting at the group-level are often forgotten in the epidemiological study of NTD infection risk [20,21], or indeed of infectious disease risk more broadly [22,23]. This apparent deficit in "contextual thinking" has occurred despite the widespread use of multi-level models, also called random effect or hierarchical models, in community-based studies of infectious disease risk in low income settings. One possible explanation for the absence of explicit contextual thinking is that the condition that necessitates the use of random effects in these multi-level models, namely the presence of within-group correlation in the outcome of interest, is almost never reported as an outcome of substantive interest in studies of NTD risk. Intra-group correlation (and the need for group-level random effects) is evidence that the grouping level chosen, be it household or geographic or administrative area, has a role in shaping variation in risk, and therefore points to the importance of group-level effects on individual infection.

A number of authors have described the value (and, it could be argued, the need [24]) of considering and reporting measures of general contextual effect, such as within-group correlation or between-group heterogeneity, from multi-level studies of disease risk [12,24–29]. Such effects are described as "general" as they refer only to influence of the cluster boundaries, rather than the specific contextual characteristics of the cluster [28]. Quantification of the extent and level at which infection risk varies between these clusters of individuals can contribute to the development of research questions that are explicitly contextual, and which therefore seek to better understand how the conditions in which people live impact upon their health [27,28]. Moreover, if health inequalities can be defined as differences in health status between groups of individuals [30], estimating general contextual effects (GCE), such as the median odds ratio or the intra-cluster correlation coefficient, can also provide a simple and standardized means with which to quantify and compare health inequalities within and between populations, and for different health outcomes [31]. Estimation of these group-level effects is straightforward to integrate into the multi-level analysis of community-based disease risk [32-35], and can provide fundamental information on the levels of variation that exist within a population.

Here, we describe a general contextual analysis that seeks to quantify the role of group-level effects in shaping variation in endemic NTD risk at a range of levels of aggregation in a rural farming community in Kenya. Since the NTDs commonly co-occur with HIV/AIDS, tuberculosis (TB) and malaria [36], we compare the GCE observed for NTDs with infection with pathogens causing these three diseases. In addition to describing the levels of variation in helminth, bacterial, protozoal and viral infection risk that exists within a single population, our aim is to use this analysis to demonstrate the value that can be added to the multi-level analysis of NTD risk through the quantification of GCE.

Methods

Data were collected as part of the 'People, Animals and their Zoonoses' (PAZ) study [37]. This was a large cross-sectional survey of all eligible and consenting members of 416 randomly selected households in a single, mixed farming community in western Kenya. In total, 2113 people of all ages meeting the inclusion criteria (\geq 5 years and without conditions that may have made blood sampling harmful) were included and sampled between September 2010 and July 2012. Samples from participants were tested for current infection with a range of pathogens. A questionnaire was conducted with all recruited participants in their preferred language (Kiswahili, Dholuo, Kiluhya or English).

Grouping levels

Sampled individuals were nested within randomly selected households. These represented family groups living within a single compound, sharing meals and a common water source.

The average reported household size was 7.6 people (range 1 to 30), from which our average sample size was 5.1 (range 1 to 21). Households were selected from within sublocations, the smallest administrative unit in Kenya. We sampled between 1 and 8 households in all 141 sublocations in the study area. The PAZ study focused on zoonotic disease risk, and the number of households selected per sublocation was proportional to the cattle population (see [37] for further details). Sublocations were nested within constituencies, the level at which government funding for development, and particularly for poverty alleviation, is allocated in Kenya. There were a total of 13 constituencies in the study area. On the basis of the 2009 census (OpenData, http://www.opendata.go.ke), sublocations in the study area had a median total population of 4,809 (range 1,187–33,352) and a median area of 10.8 km² (range 0.96–64.6). Constituencies were made up of between 3 and 7 sublocations. The geographic distribution of sampled households, sublocations and constituencies is shown in Fig 1.

Classifying outcomes

Several infectious agents, including a number that cause NTDs, were highly prevalent in the population under study. Here we focus on those with an individual-level prevalence after adjustment for the complex study design (see [37] for details) greater than 5%. These were *Ancylostoma duodenale* and/or *Necator americanus* (hereafter, hookworm) (36.3% (95% CI



Fig 1. The geographic distribution of sampled households, sublocations and constituencies (Base layers from https://gadm.org/index.html). https://doi.org/10.1371/journal.pntd.0007016.g001

32.8–39.9)); Entamoeba histolytica/dispar (30.1% (95% CI 27.5–32.8)); Plasmodium falciparum (29.4% (95% CI 26.8–32.0)); Taenia spp. (causing taeniasis) (19.7% (95% CI 16.7–22.7)); Taenia solium (causing cysticercosis) (5.8% (95% CI 4.4–7.2)); Ascaris lumbricoides (10.4% (95% CI 8.1–12.7)); Trichuris trichiura (10.0% (95% CI 8.2–11.7)); Mycobacterium tuberculosis (8.2% (95% CI 6.8–9.6)); Schistosoma mansoni (5.9% (95% CI 3.7–8.1)); and HIV (5.3% (95% CI 4.2–6.3)).

Individuals were classified as infected with *P. falciparum*, the only agent of malaria identified in the study area, if parasites were observed by light microscopy on thick or thin blood smears stained with Giemsa. Infection with the soil-transmitted helminths (hookworm, *A. lumbricoides, T. trichiura*) and *S. mansoni* was defined as the presence of at least one egg in a single faecal sample examined following preparation using the Kato-Katz (KK) [38] and formal ether concentration (FEC) techniques [39]. Infection with *E. histolytica/dispar* was defined as the presence of at least one cyst in a single faecal sample prepared using the FEC technique. *M. tuberculosis* infection was determined using a gamma-interferon assay (QuantiFERON-TB test, Cellestis) and HIV infection diagnosed using a rapid strip test (SD Bioline HIV 1/2 3.0, Standard Diagnostics). Infection with *Taenia* species (causing taeniasis, or the presence of an adult tapeworm in the gastrointestinal tract) was defined on the basis of a non-species specific copro-antigen ELISA [40], whilst cysticercosis due to *T. solium* (the presence of encysted larvae) was determined using a HP10-Ag ELISA on serum [41].

Ethical approval

Ethical approval for this study was granted by the Kenya Medical Research Institute (KEMRI) Ethical Review Board (SCC1701). All participants or their guardians provided written informed consent. Individuals found to be infected with helminths or protozoa (including *P. falciparum*) were offered treatment free of charge by study clinical officers. Referral to local health facilities was provided where necessary.

Model specification

The entire sample of 2113 people was used for the general contextual analysis. Missing-ness was present in all outcome measures and ranged from 0.05% (for *P. falciparum*) to 11.1% (for *M. tuberculosis*). Missing-ness was related to an absence of a particular sample type (blood or faeces), typically due to inadequate volumes collected or because of participant unwillingness to provide it.

Four-level logistic regression models were specified with infection as a binary outcome (infected/not infected) for each pathogen. Probability of infection was related to a set of predictors at the individual-level and random effects at the household-, sublocation- and constituency-levels. These models estimated the log odds of individual infection together with the variance at the intercept for the household (σ_H^2), sublocation (σ_{SL}^2) and constituency (σ_C^2) levels for an individual *i* living in household *j* in sublocation *k* in constituency *l*. The regression equation can be summarised as $logit(\pi_{ijkl}) = \beta_0 + \beta X + H_{0jkl} + SL_{0KL} + C_{0l}$. Our primary motivation for this analysis was to quantify general (rather than specific) contextual effects operating at each of the three grouping levels. However, age, sex, education status and ethnicity were included as fixed effects, *X*, at the individual level in order to assess the impact of withinhousehold composition on between-group variation. Models with and without fixed effects were estimated for each pathogen. A quadratic term was included for the continuous predictor age (recorded as 5 year intervals) based on the expectation of non-linear relationships with infection risk for several pathogens [37]. The continuous age variable was scaled to have a mean of zero and standard deviation of one.

Models were estimated for each pathogen in WinBUGS 1.4.3 (http://www.mre-bsu.cam.ac. uk/software/bugs/) using weakly informative normal priors for all fixed and random effects. The standard deviation for each of the group-level random effects was defined using a wide uniform hyper-prior (i.e. Uniform(1,100)). Model convergence was confirmed by visual assessment of MCMC chains. Inference was based on 3 chains that were allowed to run for at least 70,000 iterations after a burn-in of at least 30,000 with a thinning interval of at least 10. We derived the median and 2.5th and 97.5th percentiles from posterior distributions of each parameter for point estimates and 95% credibility intervals, respectively. All data manipulation was performed in R statistical environment (R version 3.1.1, http://cran.r-project.org/) with logistic regression models estimated via the *R2WinBUGs* package [42]. Estimation was performed within a Bayesian framework based on MCMC to reduce bias in the estimates for random effect parameters [43], and for ease of estimation of the associated uncertainty for GCE.

Quantifying general contextual effects

Variance partition coefficient. The variance partition co-efficient (VPC) was calculated from the outputs from each multi-level logistic regression model for each pathogen using the latent variable method [44,45]. This approach assumes that the propensity for individual infection is on a continuous scale and that only those people for which a threshold is exceeded can be considered to acquire infection. Whilst it has been suggested that such a justification is difficult to make for truly discrete outcomes [44], such as infection, interacting thresholds relating to exposure (for example, infectious dose) and susceptibility (such as immunity) could be envisaged. The unobserved latent variable (or probability of infection) is assumed to follow a logistic distribution, with variance equal to $\pi^2/3$ (i.e. 3.29). Using this approach, the VPC at the household (_H), sublocation (_{SL}) and constituency (_C) levels were [27]:

$$VPC_{H} = \frac{\sigma_{C}^{2} + \sigma_{SL}^{2} + \sigma_{H}^{2}}{\sigma_{C}^{2} + \sigma_{SL}^{2} + \sigma_{H}^{2} + \pi^{2}/3}$$
$$VPC_{SL} = \frac{\sigma_{C}^{2} + \sigma_{SL}^{2}}{\sigma_{c}^{2} + \sigma_{SL}^{2} + \sigma_{H}^{2} + \pi^{2}/3}$$
$$VPC_{c} = \frac{\sigma_{C}^{2}}{\sigma_{c}^{2} + \sigma_{SL}^{2} + \sigma_{H}^{2} + \pi^{2}/3}$$

The VPC represents the correlation in the probability of infection between two individuals randomly selected from the same household (VPC_H), sublocation (VPC_{SL}) or constituency (VPC_C). For the models described, the VPC can be considered to be equivalent to the intraclass co-efficient (ICC).

In order to further evaluate the importance of higher contextual levels in structuring variation in individual infection, we also calculated the proportion of variance (PTV) at the sublocation and household level as a fraction of total variation:

$$PTV_{H} = \frac{\sigma_{H}^{2}}{\sigma_{C}^{2} + \sigma_{SL}^{2} + \sigma_{H}^{2} + \pi^{2}/3}$$

$$PTV_{SL} = \frac{\sigma_{SL}^2}{\sigma_c^2 + \sigma_{SL}^2 + \sigma_H^2 + \pi^2/3}$$

We do not directly estimate PTV_C since this is equivalent to VPC_C .

Median odds ratio. The median odds ratio (MOR) provides a measure of heterogeneity in an individual-level outcome between groups. It represents the median value of the odds ratio comparing group-level residuals from randomly selected pairs of individuals living in a group at higher risk and those from a group at lower risk [27]. The MOR can be considered to represent the (median) difference in odds when moving between groups. It can be calculated as [26]:

$$\begin{aligned} \text{MOR}_{H} &\approx \exp(0.95 * \sqrt{\sigma_{C}^{2} + \sigma_{SL}^{2} + \sigma_{H}^{2}}) \\ \text{MOR}_{SL} &\approx \exp(0.95 * \sqrt{\sigma_{C}^{2} + \sigma_{SL}^{2}}) \\ \text{MOR}_{C} &\approx \exp(0.95 * \sqrt{\sigma_{C}^{2}}) \end{aligned}$$

Where there is little variation in individual risk between groups, the MOR will be close to one.

Spatial clustering

Geographic effects not captured in the non-spatial multi-level logistic regression models were identified by testing the standardised sublocation level residual log odds for evidence of spatial clustering in high or low values using the spatial scan statistic [46]. The default maximum cluster size of 50% of the sample was chosen using a circular spatial window. The sublocation was used as the highest contextual level for the exploration of spatial clustering due to the small number of groups at the constituency level (n = 13). We used a normal model in SatScan version 9.4.4 (www.satscan.org). To account for differences in sample sizes, the number of individuals sampled in each sublocation were included as model weights [47]. Sublocation residuals for spatial analysis were drawn from a three-level logistic regression model (with random effects for household and sublocation only) with and without adjustment for withinhousehold compositional effects.

Results

General characteristics

The variation in prevalence of each infectious agent across the range of variables included as fixed effects is shown in Table 1. Variation in prevalence of infection between self-reported members the different ethnic groups was particularly apparent, and most notably so for *A. lumbricoides*, *T. trichiura*, *Taenia* spp. (causing taeniasis) and HIV. Heterogeneity in the prevalence of infection with each of these pathogens, and with *S. mansoni* and *T. solium* (causing cysticercosis), was also evident between constituencies.

Fixed effects

Co-efficients from the adjusted models (M2) for each pathogen are shown in Table 2 (STH and *S. mansoni*), Table 3 (*E. histolytica/dispar*, *Taenia* spp. and *T. solium*) and Table 4 (HIV, *P. falciparum*, *M. tuberculosis*). Male gender was associated with increased odds of hookworm and *S. mansoni* infection, with weaker evidence for taeniasis. Females had greater odds of *T. trichiura*, *E. histolytica/dispar*, and HIV infection and *T. solium* cysticercosis. There was no evidence of a relationship between sex and *A. lumbricoides* (Table 2), *M. tuberculosis*, or *P. falciparum* (Table 4) infection. Hookworm (Table 2), *M. tuberculosis* and HIV (Table 4) infection increased with age, with evidence in each case of a negative quadratic effect. Infection declined

Table 1. Participant characteristics and percentage infected with each pathogen.

	AL	HW	TT	SM	EH	TA	TS	HIV	МТ	PF
All	8.8%	35.5%	11.2%	7.8%	29.9%	19.9%	6.6%	5.9%	8.0%	29.7%
Sex										
Female	9.3%	33.0%	13.1%	6.3%	32.3%	18.6%	7.7%	7.9%	7.6%	28.1%
Male	8.2%	38.4%	8.9%	9.7%	27.1%	21.4%	5.4%	3.6%	8.4%	31.7%
Education										
None	7.6%	46.2%	13.0%	6.5%	25.9%	19.1%	9.5%	11.3%	9.1%	16.2%
Primary	9.3%	37.4%	11.7%	8.5%	31.6%	19.6%	6.2%	6.0%	7.3%	31.9%
Beyond primary	7.4%	24.5%	8.3%	6.4%	25.6%	20.8%	6.7%	3.1%	9.8%	28.4%
Ethnicity										
Luhya	10.7%	37.6%	9.3%	9.6%	30.7%	17.7%	5.3%	4.3%	7.8%	28.2%
Luo	13.2%	28.6%	23.2%	6.8%	27.6%	32.5%	7.8%	12.3%	10.8%	29.5%
Samia	1.6%	32.0%	10.2%	9.4%	28.8%	17.3%	7.5%	5.0%	5.4%	36.0%
Teso	2.0%	41.2%	0.7%	2.0%	31.4%	11.3%	8.6%	2.6%	6.6%	30.1%
Age group										
5 to 9	13.9%	24.8%	10.9%	5.8%	26.4%	18.0%	6.5%	0.9%	1.8%	51.6%
10 to 14	10.6%	34.3%	15.1%	9.6%	31.9%	20.1%	7.1%	2.3%	3.5%	49.1%
15 to 24	8.5%	35.5%	13.4%	9.3%	35.0%	19.9%	5.5%	1.8%	8.1%	23.0%
25 to 39	6.0%	40.6%	8.5%	8.2%	28.1%	19.2%	6.2%	13.2%	14.1%	14.2%
40+	4.7%	42.7%	8.1%	6.7%	28.6%	22.1%	7.5%	11.5%	12.9%	9.0%
Constituency										
Alego Usonga	11.8%	23.1%	24.9%	10.7%	22.4 <mark>%</mark>	43.9%	17.5%	12.4%	10.5%	31.6%
Budalangi	10.4%	24.8%	24.3%	30.2%	20.8%	21.4%	3.1%	11.5%	12.5%	25.7%
Bumula	11.4%	48.5%	0.6%	2.4%	31.6%	10.3%	6.7%	1.7%	2.7%	26.8%
Butula	14.7%	47.3%	16.0%	2.7%	32.9%	24.3%	7.1%	5.8 <mark>%</mark>	6.8%	32.7%
Funyula	2.4%	32.3%	9.9%	9.6%	29.9%	17.7%	6.6%	4.2%	5.6%	36.2%
Matayos	15.3%	50.8%	7.6%	1.7%	32.2%	26.7%	4.8%	6.3%	10.3%	35.7%
Matungu	16.5%	34.1%	3.3%	7.7%	33.0%	15.4%	2.1%	3.2%	10.6%	36.2%
Mumias West	14.5%	37.3%	14.5%	7.2%	21.4%	21.4%	2.3%	1.1%	9.2%	21.6%
Nambale	3.3%	36.3%	0.9%	2.4%	41.0%	6.4%	7.0%	0.5%	7.3%	25.0%
Teso North	0.9%	37.0%	0.0%	1.9%	37.0%	10.5%	17 <mark>.6%</mark>	1.9%	5.9%	31.5%
Teso South	2.4%	44.6%	4.8%	2.4%	27.9%	12.2%	5 <mark>.1%</mark>	3.9%	4.1%	23.5%
Ugenya	10.9%	20.7%	14.1%	1.1%	27.2%	21.1%	0.0%	16.0%	15.3%	22.3%
Ugunja	16.8%	2 <mark>6.9</mark> %	24.4%	10.1%	31.4%	34.5%	1.6%	10.3%	11.7%	33.3%
Total tested	2013	2013	2013	2013	2057	1993	2092	2105	1879	2112

AL = A. lumbricoides; HW = Hookworm; TT = T. Trichiura; SM = S. mansoni; EH = E. histolytica/dispar; TA = Taenia spp.; TS = T. solium; HIV = HIV; MT = M. tuberculosis; PF = P. falciparum

https://doi.org/10.1371/journal.pntd.0007016.t001

with age for *T. trichiura*, *A. lumbricoides* (Table 2) and *P. falciparum* (Table 4). Having an education beyond primary school tended to reduce odds of infection for the majority of pathogens under study, although this was only significant in the case of hookworm (Table 2). There were strong relationships between ethnicity and infection for several pathogens, including substantially reduced odds among people of Samia and Teso ethnicity for *A. lumbricoides* compared to the Luhya baseline. Odds of *T. trichiura* infection were reduced among people of Teso ethnicity and elevated among people of Luo ethnicity when compared to the Luhya baseline. The odds of HIV infection was also higher among individuals of Luo ethnicity than the Luhya baseline.

	Hookworm		A. lumb	ricoides	T. trie	chiura	S. mansoni	
	M1	M2	M1	M2	M1	M2	M1	M2
Intercept	-0.8 (-1.1, -0.4)	-0.3 (-0.9, 0.3)	-3.4 (-4.3, -2.6)	-3.5 (-4.6, -2.4)	-3.2 (-4.3, -2.2)	-2.8 (-4.0, -1.8)	-5 (-6.5, -3.8)	-5.8 (-7.8, -4.1)
Age	-	0.5 (0.4, 0.7)*	-	-0.9 (-1.2, -0.6)*	-	-0.5 (-0.8, -0.3)*	-	0.2 (-0.1, 0.6)
Age x Age	-	-0.2 (-0.3, -0.1)*	-	0.2 (-0.0, 0.5)	-	0.1 (-0.0, 0.3)	-	-0.3 (-0.6, -0.1)*
Male gender	-	0.5 (0.2, 0.7)*	-	-0.2 (-0.6, 0.2)	-	-0.4 (-0.8, -0.1)*	-	1.2 (0.7, 1.7)*
Luo ethnicity ¹	-	-0.1 (-0.7, 0.4)	-	0.4 (-0.3, 1.2)	-	1.1 (0.4, 1.8)*	-	0.0 (-1.4, 1.3)
Samia ethnicity ¹	-	-0.2 (-0.8, 0.4)	-	-1.9 (-3.4, -0.5)*	-	0.1 (-0.7, 0.9)	-	-0.8 (-1.8, 0.3)
Teso ethnicity ¹	-	0.1 (-0.4, 0.7)	-	-1.7 (-3.1, -0.4)*	-	-2.2 (-4.3, -0.6)*	-	-1.9 (-4.0, 0.0)
Primary ²	-	-0.4 (-0.9, 0.0)	-	0.0 (-0.8, 0.8)	-	-0.3 (-0.9, 0.4)	-	0.5 (-0.5, 1.6)
Beyond primary ²	-	-1.2 (-1.7, -0.6)*	-	-0.4 (-1.3, 0.6)	-	-0.7 (-1.4, 0.1)	-	-0.3 (-1.5, 0.8)
σ^{2}_{H}	1.4 (0.9, 2)	1.5 (1.0, 2.2)	2.7 (1.6, 4.5)	3.1 (1.9, 5.0)	0.6 (0.0, 1.3)	0.4 (0.0, 1.1)	3.8 (2.0, 7.1)	5.0 (2.7, 9.4)
σ^2_{SL}	0.3 (0.0, 0.7)	0.4 (0.1, 0.8)	0.1 (0.0, 0.8)	0.1 (0.0, 0.8)	1.0 (0.5, 1.9)	1.0 (0.5, 1.9)	2.2 (0.6, 5.1)	3.1 (1.0, 6.6)
σ^2_C	0.2 (0.0, 0.8)	0.2 (0.0, 1.0)	1.2 (0.3, 4.0)	0.3 (0.0, 1.7)	2.7 (0.9, 9.0)	1.2 (0.3, 4.7)	2.6 (0.8, 9.3)	3.3 (0.9, 11.5)
VPC _H	37.0% (29.5, 45.7)	39.6% (31.7, 49.1)	56.1% (43.8, 69.6)	52.5% (40.5, 65.2)	57.0% (41.4, 76.6)	45.0% (30.5, 65.9)	73.3% (61.8, 84)	78.2% (68.1, 87.3)
VPC _{SL}	10.2% (3.2, 20.2)	11.8% (4.3, 23.5)	18.7% (6.7, 41.8)	6.8% (0.5, 23.9)	49.4% (31.4, 72.7)	38.2% (23.0, 61.6)	41.4% (23.7, 63)	44.6% (25.4, 65.3)
VPC _C	3.9% (0.4, 13.4)	4.2% (0.2, 16)	15.7% (5.0, 39.1)	4.2% (0.0, 20.7)	35.0% (15.3, 64.7)	19.9% (5.5, 50)	21.6% (7.7, 48)	22.4% (7.5, 48.8)
PTV _{HH}	26.5% (18.7, 35.3)	27.4% (19.1, 36.9)	36.2% (23.1, 49.6)	44.5% (31, 57.2)	7.1% (0.5, 16.5)	6.0% (0.0, 16.9)	31.1% (17.1, 48.1)	32.9% (18.2, 50.6)
PTV _{SL}	5.7% (0.1, 12.6)	6.8% (1.2, 14.1)	1.7% (0.0, 10.6)	1.2% (0.0, 10.2)	13.2% (5.5, 24.5)	17.1% (7.4, 29.2)	18.0% (4.8, 34.7)	20.3% (6.8, 37.7)
MOR _H	3.7 (3.1, 4.9)	4.0 (3.2, 5.4)	7 (4.6, 13.6)	6.1 (4.1, 10.6)	7.3 (4.3, 22.6)	4.8 (3.1, 11.0)	17.4 (8.9, 52.2)	26.1 (12.4, 91.6)
MOR _{SL}	2.0 (1.5, 2.8)	2.1 (1.6, 3.2)	3.1 (1.9, 7.2)	1.9 (1.2, 3.8)	6.3 (3.6, 20.6)	4.2 (2.8, 10.0)	8 <mark>.5</mark> (4.3, 28.4)	11.5 (5.2, 43.7)
MOR _C	1.5 (1.1, 2.3)	1.6 (1.1, 2.6)	2. <mark>8 (1</mark> .7, 6.7)	1.7 (1.1, 3.5)	4.7 (2.5, 17.3)	2.8 (1.7, 7.9)	4.7 (2.3, 18.3)	5.7 (2.5, 25.0)
DIC ³	2523.4	2442.9	1121.3	1079.4	1308.6	1296.8	835.9	790.5

Table 2. Posterior median estimates and 95% credibility intervals from null (M1) and adjusted (M2) models examining individual infection with hookworm, *Ascaris lumbricoides, Trichuris trichiura* and *Schistosoma mansoni*.

* 95% credibility intervals do not include 1

¹ Luhya ethnicity baseline

² No education baseline

³ Deviance information criteria (DIC) is an estimate of predictive error: the lower the better.

https://doi.org/10.1371/journal.pntd.0007016.t002

General contextual effects

The posterior distribution of household-, sublocation- and constituency-level variance, VPCs, MORs and PTVs for the gastrointestinal nematodes and *S. mansoni* are shown in Table 2, in Table 3 for *E. histolytica/dispar* and *Taenia* species, and in Table 4 for HIV, *P. falciparum* and *M. tuberculosis*. Some degree of clustering at the household-level was apparent for all pathogens. This was consistently highest for the helminth parasites (Fig 2), for which there was substantial heterogeneity in risk of infection between individuals in different households, as evidenced by MORs which exceeded 3.5 for each helminth infection in both the null and adjusted models (Table 2 and Table 3). To put these effects into context, we would expect that were an individual to permanently move from one household to another with higher risk anywhere in the study area, their odds of infection with the helminth parasites under study would change by at least 3.5 times. This household clustering effect was particularly large for *S. mansoni* (Table 2) and *T. solium* cysticercosis (Table 3). The partitioning of group-level variation was generally largest at the household-level, although the greatest proportion of individual

	E. histolyt	ica/dispar	Taeni	a spp.	T. solium		
	M1	M2	M1	M2	M1	M2	
Intercept	-1.0 (-1.2, -0.8)	-1.0 (-1.5, -0.5)	-2.1 (-2.7, -1.6)	-2.2 (-3.0, -1.4)	-4.7 (-5.7, -3.9)	-4.6 (-6.0, -3.4)	
Age	-	0.0 (-0.1, 0.2)	-	0.1 (-0.1, 0.3)	-	-0.1 (-0.4, 0.2)	
Age x Age	-	-0.1 (-0.2, 0.1)	-	0.0 (-0.2, 0.1)	-	0.2 (-0.1, 0.4)	
Male gender	-	-0.3 (-0.5, -0.1)*	-	0.3 (-0.0, 0.6)	-	-0.7 (-1.1, -0.2)*	
Luo ethnicity ¹	-	-0.2 (-0.5, 0.2)	-	0.1 (-0.6, 0.8)		0.1 (-0.9, 1.1)	
Samia ethnicity ¹	-	-0.1 (-0.6, 0.4)	-	-0.4 (-1.1, 0.4)	-	0.4 (-0.9, 1.7)	
Teso ethnicity ¹	-	0.0 (-0.4, 0.5)	-	-0.8 (-1.7, 0.0)	-	0.7 (-0.3, 1.7)	
Primary ²	-	0.3 (-0.1, 0.7)	-	0.0 (-0.5, 0.6)	-	-0.5 (-1.3, 0.3)	
Beyond primary ²	-	0.0 (-0.5, 0.5)	-	0.1 (<mark>-0.5</mark> , 0.8)	-	-0.2 (-1.1, 0.7)	
σ^2_H	0.7 (0.4, 1.1)	0.7 (0.4, 1.1)	1.6 (1.0, 2.7)	1.8 <mark>(1.1</mark> , 2.9)	1.1 (0.3, 2.5)	1.4 (0.4, 3.0)	
σ^2_{SL}	0.0 (0.0, 0.2)	0.0 (0.0, 0.2)	1.6 (0.8, 2.9)	1.6 (0.8, 2.8)	4. <mark>3 (2</mark> .4, 7.9)	4.7 (2.6, 9.0)	
σ^2_C	0.0 (0.0, 0.2)	0.0 (0.0, 0.3)	0.5 (0.1, 2.0)	0.3 (0.0, 1.5)	0.3 (0.0, 2.3)	0.3 (0.0, 2.5)	
VPC _H	19.4% (13.2, 26.8)	20.1% (13.5, 27.7)	54.0% (44.6, 64.1)	54.0% (44.6, 63.6)	64.2% (51.9, 76)	67.0% (55.2, 77.9)	
VPC _{SL}	2.5% (0.3, 8.1)	2.5% (0.2, 8.4)	30.5% (18.8, 44.8)	27.9% (16.1, 41.8)	51.7% (37. <mark>3, 6</mark> 6.3)	52.7% (37.5, 68.3)	
VPC _C	1.1% (0.0, 5.8)	1.1% (0.0, 5.9)	7.2% (0.8, 23.5)	4.2% (0.0, 18.7)	2.8% (0.0, 21)	2.6% (0.0, 20.8)	
PTV _{HH}	16.5% (10.2, 23.8)	17.2% (10.3, 24.6)	23.0% (14.1, 34)	25.7% (16.2, 37.2)	11.8% (4.0, 23.4)	13.8% (4.8, 26.4)	
PTV _{SL}	0.9% (0.0, 5.3)	0.9% (0.0, 5.8)	21.9% (11.0, 35.5)	22.4% (10.8, 35.3)	47.0% (31.0, 62.6)	48.1% (31.8, 64.3)	
MOR _H	2.3 (2.0, 2.8)	2.4 (2.0, 2.9)	6.5 (4.7, 10.0)	6.5 (4.7, 9.8)	10.1 (6.0, 21.4)	11.7 (6.8, 25.5)	
MOR _{SL}	1.4 (1.1, 1.7)	1.4 (1.1, 1.8)	4.1 (2.8, 6.6)	3.8 (2.6, 6.0)	7.9 (4.8, 16.9)	8.8 (5.1, 20.1)	
MOR _C	1.2 (1.0, 1.6)	1.2 (1.0, 1.6)	2.0 (1.3, 3.9)	1.7 (1.0, 3.3)	1.6 (1.0, 4.2)	1.6 (1.0, 4.4)	
DIC ³	2701.3	2704.9	1731.5	1731.7	838.7	822.3	

Table 3. Posterior median estimates and 95% credibility intervals from null (M1) and adjusted (M2) models examining individual infection with *Entamoeba histo-lytica/dispar*, taeniasis due to infection with *Taenia solium* or *T. saginata* and cysticercosis due to *T. solium*.

* 95% credibility intervals do not include 1

¹ Luhya ethnicity baseline

² No education baseline

³ Deviance information criteria (DIC) is an estimate of predictive error: the lower the better.

https://doi.org/10.1371/journal.pntd.0007016.t003

variation was partitioned at the constituency level (VPC_c) in null models for *T. trichiura* (Table 2) and HIV (Table 4), and the sublocation-level for *T. solium* cysticercosis (PTV_{SL}) (Table 3). Using MORs, these higher-level contextual effects could be interpreted as an almost five- and three-fold change in the odds of infection for an individual that permanently moves to a higher risk constituency for *T. trichiura* and HIV, respectively. Similarly, the median odds of an individual permanently moving to a higher risk sublocation could be expected to increase by around eight times for *T. solium* cysticercosis. Control for individual-level fixed effects resulted in declines in within-constituency correlation (VPC_C) and between-constituency heterogeneity (MOR_C) for infection with several of the pathogens under study, most notably for *A. lumbricoides* and *T. trichiura* (Table 2) and HIV (Table 4).

Spatial clustering

The spatial distribution of sublocations with evidence for clustering in high or low values of residual log odds of infection is shown in Fig 3. Large spatial clusters of both high and low values were observed from null models for *T. trichiura*, *S. mansoni*, *A. lumbricoides*, and *Taenia* spp.. There was substantial overlap in clusters for all of these pathogens and a large cluster of sublocations with elevated risk of individual HIV infection. We found no evidence of spatial structuring in the sublocation-level residual log odds of infection with *M. tuberculosis* or *T*.

	Н	IV	P. falc	iparum	M. tuberculosis		
	M1	M2	M1	M2	M1	M2	
Intercept	-3.3 (-4.1, -2.6)	-2.8 (-3.8, -1.9)	-1 (-1.2, -0.8)	-1.8 (-2.4, -1.2)	-2.6 (-3.0, -2.3)	-2.9 (-3.8, -2.2)	
Age	-	1.7 (1.3, 2.1)*	-	-1.5 (-1.7, -1.3)*	-	1.1 (0.8, 1.4)*	
Age x Age	-	-0.9 (-1.2, -0.7)*	-	0.4 (0.3, 0.6)*	-	-0.4 (-0.6, -0.3)*	
Male gender	-	-0.5 (-1.0, -0.1)*	-	0.1 (-0.1, 0.3)	-	0.3 (-0.1, 0.6)	
Luo ethnicity ¹	-	1.1 (0.4, 1.8)*	-	0.2 (-0.2, 0.6)	-	0.3 (-0.3, 0.8)	
Samia ethnicity ¹	-	0.6 (-0.3, 1.6)	-	0.2 (-0.4, 0.7)	-	-0.2 (-1.1, 0.7)	
Teso ethnicity ¹	-	-0.1 (-1.2, 0.9)	-	0.0 (-0.4, 0.5)	-	0.1 (-0.6, 0.9)	
Primary ²	-	-0.3 (-1.0, 0.4)	-	0.1 (-0.4, 0.7)	-	0.2 (-0.4, 0.9)	
Beyond primary ²	-	-0.7 (-1.6, 0.1)	-	-0.2 <mark>(-0</mark> .7, 0.4)	-	0.6 (-0.2, 1.3)	
σ_{H}^{2}	0.3 (0.0, 1.3)	0.4 (0.0, 1.7)	0.3 (0.1, 0.6)	0. <mark>4 (0.</mark> 1, 0.8)	0.2 (0.0, 0.9)	0.4 (0.0, 1.2)	
σ^2_{SL}	0.3 (0.0, 0.8)	0.3 (0.0, 1.0)	0.2 (0.0, 0.3)	0.2 (0.0, 0.4)	0 <mark>.0 (</mark> 0.0, 0.3)	0.1 (0.0, 0.4)	
σ^2_C	1.0 (0.3, 3.5)	0.6 (0.1, 2.3)	0.0 (0.0, 0.2)	0.0 (0.0, 0.3)	0.1 (0.0, 0.6)	0.2 (0.0, 0.9)	
VPC _H	34.7% (19, 56.9)	30.6% (14.0, 52.1)	12.7% (7.1, 19.7)	17.6% (10.3, 26.2)	13.5% (3.6, 27.7)	18.7% (5.7, 35.4)	
VPC _{SL}	26.8% (11.8, 51.7)	20.2% (6.6, 43.2)	4.9% (1.2, 10.3)	6.4% (1.5, 13.5)	5.9% (0. <mark>7, 1</mark> 7.6)	6.8% (0.8, 21.5)	
VPC _C	20.5% (7.7, 47.6)	12.3% (2.1, 35.9)	0.5% (0.0, 3.9)	1.0% (0.0, 6.7)	3.6% (0.1, 14.7)	4.1% (0.0, 18.8)	
РТV _{нн}	6.2% (0.1, 22.5)	8.8% (0.1, 27.9)	7.6% (2.1, 14.2)	10.8% (3.9, 19.1)	6.4% (0.1, 19.6)	10.8% (0.3, 24.8)	
PTV _{SL}	4.9% (0.1, 16)	6.4% (0.1, 19.8)	4.0% (0.6, 8.8)	4.9% (0.6, 10.8)	1.2% (0.0, 8.8)	1.5% (0.0, 9.9)	
MOR _H	3.5 (2.3, 7.3)	3.1 (2.0, 6)	1.9 (1.6, 2.3)	2.2 (1.8, 2.8)	2 (1.4, 2.9)	2.3 (1.5, 3.6)	
MOR _{SL}	3.0 (2.0, 6.5)	2.5 (1.6, 4.9)	1.5 (1.2, 1.8)	1.6 (1.3, 2.1)	1.6 (1.2, 2.3)	1.6 (1.2, 2.6)	
MOR _C	2.6 (1.7, 6)	2.1 (1.3, 4.3)	1.1 (1.0, 1.5)	1.2 (1.0, 1.6)	1.4 (1.0, 2.1)	1.5 (1.0, 2.5)	
DIC ³	1101.5	1036.2	1275.5	1276.8	2883	2545.8	

Table 4. Posterior median estimates and 95% credibility intervals from null (M1) and adjusted (M2) models examining individual infection with HIV, Plasmodium falciparum, and *Mycobacterium tuberculosis*.

* 95% credibility intervals do not include 1

¹ Luhya ethnicity baseline

² No education baseline

³ Deviance information criteria (DIC) is an estimate of predictive error: the lower the better

https://doi.org/10.1371/journal.pntd.0007016.t004

solium and relatively small clusters for *P. falciparum*, hookworm and *E. histolytica/dispar* (Fig 3). The spatial extent of the clusters of both high and low sublocation residual log odds was reduced when controlling for individual-level fixed effects in the case of HIV. Adjustment for these fixed effects resulted in a loss of significance in spatial clusters of both high and low values from the model for *A. lumbricoides*, and of high values for *T. trichiura*. Only the spatial cluster of positive sublocation residual log odds remained significant in the case of *S. mansoni* (Fig 3).

Discussion

In this general contextual analysis, we demonstrate the value of summarizing variation in individual infectious disease risk at one or more biologically relevant grouping levels using the outputs from multi-level regression. Deriving statistics such as the MOR and VPC (or ICC) as part of an exploratory analysis of infectious disease risk is straightforward, and can contribute important information about the heterogeneity that underlies population-level averages, such as prevalence [26–28,33]. Using this approach, we show that variation in individual infection risk is partitioned at the household, sublocation and constituency-levels for a range of NTDs in a rural population in Kenya. These findings point to the importance of social and/or environmental contextual conditions in shaping infection at each of these levels, and which may



Fig 2. Posterior distributions of variance partition coefficients (VPC) for each pathogen at the household (dark grey), sublocation (grey) and constituency levels (light grey) without control for individual-level predictors (SM = S. *mansoni*; TS = T. *solium*; TA = T *aenia* spp.; AL = A. *lumbricoides*; TT = T. *Trichiura*; HW = Hookworm; HIV = HIV; EH = E. *histolytica/dispar*; MA = P. *falciparum*; TB = M. *tuberculosis*).

https://doi.org/10.1371/journal.pntd.0007016.g002

provide actionable targets for public health interventions seeking to reduce both the prevalence of infection and the health inequalities observed.

An important limitation that should be recognised when interpreting these findings, and particularly when making comparisons between pathogens, is the lack of precision in many of our estimates of GCE, particularly at higher contextual levels. Hence, whilst estimates of VPC and MOR at the constituency-level were substantially different between, for example, hookworm and *S. mansoni* infection, the 95% credibility intervals overlap. This is a limitation of the sample available, both in terms of number of individuals and number of individual groups at the higher contextual levels. The magnitude of the MOR or VPC provides useful information on the importance of a particular level in structuring risk [28], and for the example of hookworm and *S. mansoni*, strongly suggests contextual drivers operating at the constituency level are more important for the latter than the former. However, when interpreting differences between pathogens at these higher contextual levels, or between different contextual levels for the same pathogen, it should be noted that the statistical support for many of the differences we observed was often limited.

A general contextual analysis can provide a tool for exploring the levels at which pathogen transmission occurs within a population [16]. For example, we show that the majority of variation in individual hookworm infection was partitioned at the household level, with comparatively smaller amounts at sublocation and constituency levels. This suggests clustering at higher contextual levels is less important for this parasite in this population than for the other STHs. Individual infection with *A. lumbricoides*, for example, was partitioned at both the household- and constituency-levels, and therefore household clusters of infection can also be considered to cluster by constituency. Household clustering was less important for *T. trichiura*, but there was substantial variation in infection between constituencies, and to a lesser extent between sublocations within constituencies. Understanding these patterns of partitioning in



Fig 3. Clusters of significantly elevated (red) and reduced (blue) sublocation level standardised residual log odds of infection for: a. Hookworm; b. *A. lumbricoides*; c. *T. trichiura*; d. *S. mansoni*; e. *E. histolytica/dispar*; f. *Taenia* spp.; g. HIV; h. *P. falciparum*. Light and dark shades of red and blue represent significant clusters from the null and adjusted logistic regression models, respectively.

https://doi.org/10.1371/journal.pntd.0007016.g003

infection risk may assist in the design of interventions that seek to reduce both the prevalence and health inequalities observed. For pathogens with limited evidence for higher level GCE, such as hookworm or *E. histolytica/dispar*, it is likely that households in all parts of the study area would need to be targeted. Interventions in high risk constituencies are likely to be more cost effective for *T. trichiura*, *A. lumbricoides* and *S. mansoni*, potentially including a focus in high risk sublocations for the latter two pathogens. The general contextual analysis approach described here could be particularly valuable in monitoring the effectiveness of an intervention, such as mass drug administration. For example, a decline in population-level prevalence but persistence of, or increase in, general contextual effects at particular grouping-levels would point to ongoing or new health inequalities. Moreover, such a finding would suggest the presence of hotspots of transmission that may impact elimination [48]. Wider usage of general contextual analysis in the study of NTD risk could therefore contribute to the post-2020 NTD roadmap that sees a transition from monitoring programme coverage to measuring impact [49].

Clustering in *T. solium* cysticercosis and *Taenia* spp. taeniasis was observed at both the household and sublocation levels. This was particularly large at the sublocation level for *T. solium* cysticercosis, but not between constituencies. Hence, while spatially heterogeneous factors appear to influence cysticercosis risk, these effects are likely to operate at small spatial scales (i.e. at the sublocation-level). Cases of human cysticercosis commonly cluster around

human tapeworm carriers [50], and Okello *et al* [51] reported hyper-endemic hotspots for *T*. *solium* infection in Lao PDR. The importance of non-spatially-structured sublocation effects in our own study area could therefore be hypothesised to reflect small-scale differences in pork consumption practices, or the existence of slaughterhouses in particular sublocations with inadequate meat inspection practices. Sublocation-level residuals for taeniasis showed substantial spatial structuring on the basis of the spatial scan statistic, and the lack of a similar finding for cysticercosis may point to a preponderance of the beef tapeworm, *T. saginata* (which does not cause human cysticercosis) over *T. solium* in the study area.

The nesting of variation in individual HIV infection at the constituency level supports the growing recognition that HIV epidemiology can be characterized as a number of diverse epidemics, often with substantial variation in prevalence even at small spatial scales [52,53]. In this part of western Kenya, individual risk of HIV infection was most concentrated in constituencies in the south-western part of the study area. Further work is needed to explore the important clustering observed, including the compositional effect of ethnicity; the Luo community who, as a group, have been previously been described to be heavily burdened by HIV [54], reside primarily in the southern part of the study area [37]. Schistosoma haematobium, which we did not test for but which is known to be an important co-factor for HIV infection in sub Saharan Africa [55], is also likely to be common in the swampy area around Lake Victoria [56], and may also contribute to the clustering observed. There were substantial overlaps in the spatial distribution of HIV infection risk and that for several NTDs, most notably S. mansoni, A. lumbricoides and T. trichiura. This supports earlier analysis of the same data that showed overlapping spatial clustering in household-level infection with these pathogens [37]. The observed co-distribution of these pathogens may point to the existence of shared environmental, cultural, behavioural or social conditions leading to poly-parasitism [19]. Alternatively, it may suggest immunological interactions between HIV and these helminth parasites that influence transmission dynamics, a hypothesis supported by a growing number of field and laboratory based studies [57].

Interestingly, between-group levels of variation were considerably lower for *P. falciparum* and *M. tuberculosis* than for any of the NTDs, with the exception of infection with *E. histoly-tica/dispar*. Previous studies on *M. tuberculosis* have suggested that the majority (>80%) of transmission events for the pathogen occurs in the public (or community) rather than domestic domain [58-60]. The comparatively small levels of individual variation partitioned at the household-level (particularly compared to the helminth pathogens under study) provides further support for these findings. Moreover, in the absence of higher level GCEs, we show there is little variation in community-level transmission between different parts of the study area for *M. tuberculosis*. Although we found evidence for a small cluster of sublocations with reduced risk of *P. falciparum* infection, the absence of higher-level contextual effects (at the sublocation- and constituency-level) for this pathogen suggests geographic or administrative place of residence does not have a major influence on infection risk. This is supported by a recent study from neighbouring Eastern Uganda which, using highly sensitive molecular-based diagnostic tests, demonstrated that the vast majority of community residents, regardless of age, demography and geographic location, were infected with malaria parasites [61].

We have explored only a limited set of fixed effects at the individual level in this analysis, and no specific contextual effects (i.e. predictors operating at group-level). Having demonstrated the importance of these grouping-levels in structuring infectious disease risk, the next analytical step would be to integrate specific contextual effects, including household, sublocation and constituency-level indicators of social or environmental conditions that may explain the variation observed. The inclusion of individual-level predictors resulted in substantial decreases in the variation at higher contextual levels for pathogens such as *A. lumbricoides*, *T.*

trichiura and HIV. There were large, overlapping spatial clusters for each of these pathogens, the size of which was reduced or made to be non-significant following the inclusion of individual level predictors. All of these pathogens had strong relationships with ethnicity, which is known to be highly spatially structured in the study area [37]. Disentangling the importance of individual level cultural and behavioural practices and local social and environmental conditions would therefore help to better understand the general contextual effects observed.

Conclusion

Quantification of general contextual effects provides a means to evaluate the importance of social and environmental conditions in structuring infectious disease risk within a population. Such an approach encourages the explicit consideration of group-level, contextual effects on individual health and can form the basis for subsequent analyses that seek to explain the variation observed. Using a general contextual analysis, we have demonstrated the existence of important place-based contextual effects for a range of pathogens in a rural farming community in Kenya and show that these are particularly large for the NTDs and HIV. This study provides evidence for important variation in infectious disease risk in this underprivileged population that point to the existence of health inequalities at a range of grouping-levels.

Supporting information

S1 Checklist. Strobe checklist. (DOC)

Acknowledgments

We are grateful for the hard work of field and laboratory teams in Busia and Nairobi, in particular James Akoko, Omoto Lazarus, Jenipher Ambaka, Fredrick Opinya, Lorren Alumasa, Daniel Cheruiyot, Alice Kiyong'a, Velma Kivali, George Omondi, Gideon Mwali, John Mwaniki, Hannah Kariuki, Lilian Achola and Maseno Cleophas.

Author Contributions

Conceptualization: William A. de Glanville, Barend M. de C. Bronsvoort, Eric M. Fèvre.

Data curation: William A. de Glanville, Lian F. Thomas.

Formal analysis: William A. de Glanville.

Funding acquisition: Eric M. Fèvre.

Investigation: William A. de Glanville, Lian F. Thomas, Elizabeth A. J. Cook, Eric M. Fèvre.

Methodology: William A. de Glanville, Lian F. Thomas, Barend M. de C. Bronsvoort, Eric M. Fèvre.

Supervision: Barend M, de C. Bronsvoort, Claire N. Wamae, Samuel Kariuki, Eric M. Fèvre.

Writing – original draft: William A. de Glanville.

Writing – review & editing: William A. de Glanville, Lian F. Thomas, Elizabeth A. J. Cook, Barend M. de C. Bronsvoort, Nicola Wardrop, Claire N. Wamae, Samuel Kariuki, Eric M. Fèvre.

References

- Petney TN, Andrews RH. Multiparasite communities in animals and humans: Frequency, structure and pathogenic significance. Int J Parasitol. Pergamon; 1998; 28: 377–393. https://doi.org/10.1016/S0020-7519(97)00189-6 PMID: 9559357
- Buck AA, Anderson RI, MacRae AA. Epidemiology of poly-parasitism. I. Occurrence, frequency and distribution of multiple infections in rural communities in Chad, Peru, Afghanistan, and Zaire. Tropenmedizin und Parasitologie. 1978; 29: 61–70. PMID: 644660
- 3. WHO. Global Report for Research on Infectious Diseases of Poverty. Geneva: World Health Organization; 2012 Aug.
- Fox MP, Rosen S, MacLeod WB, Wasunna M, Bii M, Foglia G, et al. The impact of HIV/AIDS on labour productivity in Kenya. Trop Med Int Health. 2004; 9: 318–324. <u>https://doi.org/10.1111/j.1365-3156.</u> 2004.01207.x PMID: 14996359
- Deribew A, Tesfaye M, Hailmichael Y, Negussu N, Daba S, Wogi A, et al. Tuberculosis and HIV coinfection: its impact on quality of life. Health Qual Life Outcomes. 2009; 7: 105. <u>https://doi.org/10.1186/</u> 1477-7525-7-105 PMID: 20040090
- Fürst T, Silué KD, Ouattara M, N'Goran DN, Adiossan LG, N'Guessan Y, et al. Schistosomiasis, Soil-Transmitted Helminthiasis, and Sociodemographic Factors Influence Quality of Life of Adults in Côte d'Ivoire. PLoS Negl Trop Dis. 2012; 6: e1855. https://doi.org/10.1371/journal.pntd.0001855 PMID: 23056662
- 7. Mathers CD, Loncar D. Projections of Global Mortality and Burden of Disease from 2002 to 2030. PLoS Med. 2006; 3: e442. https://doi.org/10.1371/journal.pmed.0030442 PMID: 17132052
- 8. Rose G. Sick individuals and sick populations. Int J Epidemiol. 1985; 14: 32–38. PMID: 3872850
- 9. Diez-Roux AV. Bringing context back into epidemiology: variables and fallacies in multilevel analysis. Am J Public Health. 1998; 88: 216–222. https://doi.org/10.2105/AJPH.88.2,216 PMID: 9491010
- Diez Roux AV, Aiello AE. Multilevel Analysis of Infectious Diseases. J Infect Dis. 2005; 191: S25–S33. https://doi.org/10.1086/425288 PMID: 15627228
- Allotey P, Reidpath DD, Ghalib H, Pagnoni F, Skelly WC. Efficacious, effective, and embedded interventions: Implementation research in infectious disease control. BMC Public Health. 2008; 8: 151. https://doi.org/10.1186/1471-2458-8-151
- 12. Rodriguez G, Goldman N. An assessment of estimation procedures for multilevel models with binary responses. J R Statistic Soc A. 1995; 158: 73–89.
- Haswell-Elkins M, Elkins D, Anderson RM. The influence of individual, social group and household factors on the distribution of Ascaris lumbricoides within a community and implications for control strategies. Parasitology. 1989; 98: 125–134. PMID: 2717212
- 14. Forrester JE, Scott ME, Bundy DA, Golden MH. Clustering of Ascaris lumbricoides and Trichuris trichiura infections within households. Trans R Soc Trop Med Hyg. 1988; 82: 282–288. PMID: 3188157
- Bethony J, Bethony J, Williams JT, Williams JT, Kloos H, Kloos H, et al. Exposure to Schistosoma mansoni infection in a rural area in Brazil. II: Household risk factors. Trop Med Int Health. 2001; 6: 136–145. https://doi.org/10.1046/j.1365-3156.2001.00685.x PMID: 11251910
- Cairncross S, Blumenthal U, Kolsky P, Moraes L, Tayeh A. The public and domestic domains in the transmission of disease. Trop Med Int Health. 1996; 1: 27–34. PMID: 8673819
- Walker M, Hall A, Basanez M-G. Individual Predisposition, Household Clustering and Risk Factors for Human Infection with Ascaris lumbricoides: New Epidemiological Insights. PLoS Negl Trop Dis. 2011; 5: e1047. https://doi.org/10.1371/journal.pntd.0001047 PMID: 21541362
- Lescano AG, García HH, Gilman RH, Gavidia CM, Tsang VCW, Rodriguez S, et al. Taenia solium Cysticercosis Hotspots Surrounding Tapeworm Carriers: Clustering on Human Seroprevalence but Not on Seizures. PLoS Negl Trop Dis. 2009; 3: e371. https://doi.org/10.1371/journal.pntd.0000371 PMID: 19172178
- Gazzinelli A, Correa-Oliveira R, Yang G-J, Boatin BA, Kloos H. A research agenda for helminth diseases of humans: social ecology, environmental determinants, and health systems. Garba A, editor. PLoS Negl Trop Dis. Public Library of Science; 2012; 6: e1603. <u>https://doi.org/10.1371/journal.pntd.</u> 0001603 PMID: 22545168
- 20. Armah FA, Quansah R, Luginaah I, Chuenpagdee R, Hambati H, Campbell G. Historical Perspective and Risk of Multiple Neglected Tropical Diseases in Coastal Tanzania: Compositional and Contextual Determinants of Disease Risk. PLoS Negl Trop Dis. 2015; 9: e0003939. <u>https://doi.org/10.1371/journal.pntd.0003939 PMID: 26241050</u>
- 21. Spiegel JM. Looking beyond the lamp post: addressing social determinants of neglected tropical diseases in devising integrated control strategies. The Causes and Impacts of Neglected Tropical and

Zoonotic Diseases: Opportunities for Integrated Intervention Stategies. National Academies Press; 2011. pp. 490–505.

- Noppert GA, Kubale JT, Wilson ML. Analyses of infectious disease patterns and drivers largely lack insights from social epidemiology: contemporary patterns and future opportunities. J Epidemiol Community Health. 2017; 71: 350–355. https://doi.org/10.1136/jech-2016-207967 PMID: 27799618
- Cohen JM, Wilson ML, Aiello AE. Analysis of social epidemiology research on infectious diseases: historical patterns and future opportunities. J Epidemiol Community Health. 2007; 61: 1021–1027. https:// doi.org/10.1136/jech.2006.057216 PMID: 18000122
- Merlo J, Wagner P, Austin PC, Subramanian SV, Leckie G. General and specific contextual effects in multilevel regression analyses and their paradoxical relationship: A conceptual tutorial. SSM—Population Health. 2018; 5: 33–37. https://doi.org/10.1016/j.ssmph.2018.05.006 PMID: 29892693
- 25. Merlo J. Multilevel analytical approaches in social epidemiology: measures of health variation compared with traditional measures of association. J Epidemiol Community Health. 2003; 57: 550–552. https://doi.org/10.1136/jech.57.8.550 PMID: 12883048
- Larsen K, Merlo J. Appropriate assessment of neighborhood effects on individual health: integrating random and fixed effects in multilevel logistic regression. Am J Epidemiol. 2005; 161: 81–88. <u>https://doi.org/10.1093/aje/kwi017 PMID: 15615918</u>
- Merlo J, Viciana-Fernández FJ, Ramiro-Fariñas D, Research Group of Longitudinal Database of Andalusian Population (LDAP). Bringing the individual back to small-area variation studies: a multilevel analysis of all-cause mortality in Andalusia, Spain. Soc Sci Med. 2012; 75: 1477–1487. https://doi.org/10. 1016/j.socscimed.2012.06.004.PMID: 22795359
- Merlo J, Ohlsson H, Lynch KF, Chaix B, Subramanian SV. Individual and collective bodies: using measures of variance and association in contextual epidemiology. J Epidemiol Community Health. 2009; 63: 1043–1048. https://doi.org/10.1136/jech.2009.088310 PMID: 19666637
- Petronis KR, Anthony JC. A different kind of contextual effect: geographical clustering of cocaine incidence in the USA. J Epidemiol Community Health. 2003; 57: 893–900. https://doi.org/10.1136/jech.57. 11.893 PMID: 14600117
- Kawachi I, Subramanian SV, Almeida-Filho N. A glossary for health inequalities. J Epidemiol Community Health. 2002; 56: 647–652. https://doi.org/10.1136/jech.56.9.647 PMID: 12177079
- Merlo J, Asplund K, Lynch J, Råstam L, Dobson A. Population effects on individual systolic blood pressure: A multilevel analysis of the World Health Organization MONICA Project. Am J Epidemiol. 2004; 159: 1168–1179. https://doi.org/10.1093/aje/kwh160 PMID: 15191934
- Merlo J, Chaix B, Yang M, Lynch J, Råstam L. A brief conceptual tutorial of multilevel analysis in social epidemiology: linking the statistical concept of clustering to the idea of contextual phenomenon. J Epidemiol Community Health. 2005; 59: 443–449. https://doi.org/10.1136/jech.2004.023473 PMID: 15911637
- Merlo J, Chaix B, Ohlsson H, Beckman A, Johnell K, Hjerpe P, et al. A brief conceptual tutorial of multilevel analysis in social epidemiology: using measures of clustering in multilevel logistic regression to investigate contextual phenomena. J Epidemiol Community Health. 2006; 60: 290–297. https://doi.org/ 10.1136/jech.2004.029454 PMID: 16537344
- Merlo J, Chaix B, Yang M, Lynch J, Råstam L. A brief conceptual tutorial on multilevel analysis in social epidemiology: interpreting neighbourhood differences and the effect of neighbourhood characteristics on individual health. J Epidemiol Community Health. 2005; 59: 1022–1028. https://doi.org/10.1136/ jech.2004.028035 PMID: 16286487
- **35.** Merlo J, Yang M, Chaix B, Lynch J, Råstam L. A brief conceptual tutorial on multilevel analysis in social epidemiology: Investigating contextual phenomena in different groups of people. J Epidemiol Community Health. BMJ Publishing Group Ltd; 2005; 59: 729–736. https://doi.org/10.1136/jech.2004.023929 PMID: 16100308
- 36. Hotez PJ, Molyneux DH, Fenwick A, Ottesen E, Sachs SE, Sachs JD. Incorporating a rapid-impact package for neglected tropical diseases with programs for HIV/AIDS, tuberculosis, and malaria: A comprehensive pro-poor health policy and strategy for the developing world. PLoS Med. 2006; 3: 576–584. https://doi.org/10.1371/journal.pmed.0030102 PMID: 16435908
- Fevre EM, de Glanville WA, Thomas LF, Cook EAJ, Kariuki S, Wamae CN. An integrated study of human and animal infectious disease in the Lake Victoria crescent small-holder crop-livestock production system, Kenya. BMC Infect Dis. 2017; 17: 342. https://doi.org/10.1186/s12879-017-2441-6
- Katz N, Chaves A, Pellegrino J. A simple device for quantitative stool thick-smear technique in Schistosomiasis mansoni. Rev Inst Med Trop Sao Paulo. 1972; 14: 397–400. PMID: 4675644
- Allen AV, Ridley DS. Further observations on the formol-ether concentration technique for faecal parasites. J Clin Pathol. 1970; 23: 545–546. PMID: 5529256

- Allan JC, Velasquez-Tohom M, Torres-Alvarez R, Yurrita P, Garcia-Noval J. Field trial of the coproantigen-based diagnosis of Taenia solium taeniasis by enzyme-linked immunosorbent assay. Am J Trop Med Hyg. 1996; 54: 352–356. PMID: 8615446
- Harrison LJS, Joshua GWP, Wright SH, Parkhouse RME. Specific detection of circulating surface/ secreted glycoproteins of viable cysticerci in Taenia saginata cysticercosis. Parasite Immunology. 1989; 11: 351–370. https://doi.org/10.1111/j.1365-3024.1989.tb00673.x PMID: 2674862
- 42. Sturtz S, Ligges U, Gelman A. R2WinBUGS: A Package for Running WinBUGS from R. J Stat Softw. 2005; 12.
- **43.** Browne WJ, Draper D. A comparison of Bayesian and likelihood-based methods for fitting multilevel models. Bayesian Analysis. 1st ed.: 473–514.
- 44. Snijders TAB, Bosker RJ. Multilevel Analysis. SAGE; 2011.
- Austin PC, Merlo J. Intermediate and advanced topics in multilevel logistic regression analysis. Stat Med. 2017; 36: 3257–3277. https://doi.org/10.1002/sim.7336 PMID: 28543517
- 46. Kulldorff M. A spatial scan statistic. Commun Stat. 2007; 26: 1481–1496. https://doi.org/10.1080/ 03610929708831995
- Alton GD, Pearl DL, Bateman KG, McNab B, Berke O. Comparison of covariate adjustment methods using space-time scan statistics for food animal syndromic surveillance. BMC Vet Res. 2013; 9: 231. https://doi.org/10.1186/1746-6148-9-231 PMID: 24246040
- Harris JR, Wiegand RE. Detecting infection hotspots: Modeling the surveillance challenge for elimination of lymphatic filariasis. PLoS Negl Trop Dis. 2017; 11: e0005610. <u>https://doi.org/10.1371/journal.pntd.0005610 PMID: 28542274</u>
- Lim MD, Brooker SJ, Belizario VY, Gay-Andrieu F, Gilleard J, Levecke B, et al. Diagnostic tools for soiltransmitted helminths control and elimination programs: A pathway for diagnostic product development. PLoS Negl Trop Dis. 2018; 12: e0006213. https://doi.org/10.1371/journal.pntd.0006213 PMID: 29494581
- Lescano AG, García HH, Gilman RH, Gavidia CM, Tsang VCW, Rodriguez S, et al. Taenia solium cysticercosis hotspots surrounding tapeworm carriers: clustering on human seroprevalence but not on seizures. PLoS Negl Trop Dis. 2009; 3: e371. https://doi.org/10.1371/journal.pntd.0000371 PMID: 19172178
- Okello A, Ash A, Keokhamphet C, Hobbs E, Khamlome B, Dorny P, et al. Investigating a hyper-endemic focus of Taenia solium in northern Lao PDR. Parasit Vectors. 2014; 7: 134. https://doi.org/10.1186/1756-3305-7-134 PMID: 24678662
- Feldacker C, Ennett ST, Speizer I. It's not just who you are but where you live: an exploration of community influences on individual HIV status in rural Malawi. Soc Sci Med. 2011; 72: 717–725. https://doi.org/ 10.1016/j.socscimed.2011.01.003 PMID: 21316134
- 53. Wilson D, Halperin DT. "Know your epidemic, know your response": a useful approach, if we get it right. Lancet. 2008; 372: 423–426. https://doi.org/10.1016/S0140-6736(08)60883-1 PMID: 18687462
- Ayikukwei R, Ngare D, Sidle J, Ayuku D, Baliddawa J, Greene J. HIV/AIDS and cultural practices in western Kenya: the impact of sexual cleansing rituals on sexual behaviours. Cult Health Sex. 2008; 10: 587–599. https://doi.org/10.1080/13691050802012601 PMID: 18649197
- Ndeffo Mbah ML, Poolman EM, Drain PK, Coffee MP, van der Werf MJ, Galvani AP. HIV and Schistosoma haematobium prevalences correlate in sub-Saharan Africa. Trop Med Int Health. 2013; 18: 1174– 1179. https://doi.org/10.1111/tmi.12165 PMID: 23952297
- 56. Sang HC, Muchiri G, Ombok M, Odiere MR, Mwinzi PNM. Schistosoma haematobium hotspots in south Nyanza, western Kenya: prevalence, distribution and co-endemicity with Schistosoma mansoni and soil-transmitted helminths. Parasit Vectors. 2014; 7: 125. <u>https://doi.org/10.1186/1756-3305-7-125</u> PMID: 24667030
- Webb EL, Ekii AO, Pala P. Epidemiology and immunology of helminth-HIV interactions. Curr Opin HIV AIDS. 2012; 7: 245–253. https://doi.org/10.1097/COH.0b013e32835210cd PMID: 22411451
- Verver S, Warren RM, Munch Z, Richardson M, van der Spuy GD, Borgdorff MW, et al. Proportion of tuberculosis transmission that takes place in households in a high-incidence area. Lancet. 2004; 363: 212–214. https://doi.org/10.1016/S0140-6736(03)15332-9 PMID: 14738796
- Martinez L, Shen Y, Mupere E, Kizza A, Hill PC, Whalen CC. Transmission of Mycobacterium Tuberculosis in Households and the Community: A Systematic Review and Meta-Analysis. Am J Epidemiol. 2017; 185: 1327–1339. https://doi.org/10.1093/aje/kwx025 PMID: 28982226
- Middelkoop K, Mathema B, Myer L, Shashkina E, Whitelaw A, Kaplan G, et al. Transmission of tuberculosis in a South African community with a high prevalence of HIV infection. J Infect Dis. 2015; 211: 53–61. https://doi.org/10.1093/infdis/jiu403 PMID: 25053739

 Katrak S, Day N, Ssemmondo E, Kwarisiima D, Midekisa A, Greenhouse B, et al. Community-wide Prevalence of Malaria Parasitemia in HIV-Infected and Uninfected Populations in a High-Transmission Setting in Uganda. J Infect Dis. 2016; 213: 1971–1978. https://doi.org/10.1093/infdis/jiw057 PMID: 26908725